Author's response to reviews

Title: Application of Ni-Ti Alloy connector for the treatment of comminuted coronal plane supracondylar-condylar femoral fractures

Authors:

Yuntong Zhang (mone1030@126.com)
Xue Zhao (mone1030@sina.cn)
Yang Tang (mone1030@sina.com)
Chuncai Zhang (mone1030@163.com)
Shuogui Xu (zone1030@gmail.com)
Yang Xie (mone1030@yeah.net)

Version: 5 Date: 19 September 2013

Author's response to reviews: see over
We would like to express our sincere thanks to the editor and reviewer for the constructive and positive comments. The quality of written English has been checked and improved by authors and a native meticulously. We have incorporated the comments and suggestions from the reviewer into our revised manuscript and we have responded to each of the concerns and comments from the reviewer on a point-by-point basis below. All changes have been made using the "Track Changes" in revised manuscript.

**Reviewer's report**
**Title:** Application of Ni-Ti Alloy connector for the treatment of comminuted coronal plane supracondylar-condylar femoral fractures
**Version:** 4 **Date:** 23 August 2013
**Reviewer:** Rajesh K Rohilla

**Reviewer's report:**
Needs grammatical corrections

All comments below are major compulsory revisions

**Reviewer:** Line 98. Eight fractures involved the lateral condyle and three involved the medial condyle. What fracture occurred in rest of ten patients.
**Authors:** We are sorry for this oversight. The sentence should have read “Eight fractures involved the lateral condyle and thirteen involved the medial condyle.” It has been changed in the updated manuscript (line 101).

**Reviewer:** Line 99. Five patients had fractures in the right knee 6 patients had fractures of the left Which side was involved in rest of ten patients?
**Authors:** The sentence should have read that 16 patients had fractures in the right knee, not 6 as originally written. We have made the appropriate change in the updated manuscript (line 102).

**Level of interest:** An article of importance in its field
**Quality of written English:** Needs some language corrections before being published
**Statistical review:** No, the manuscript does not need to be seen by a statistician.
**Declaration of competing interests:** I declare that I have no competing interests' below